Author's response to reviews

Title: Effects of kolaviron on hepatic oxidative stress in streptozotocin induced diabetes

Authors:

Oluwafemi O Oguntibeju (oguntibejuo@cput.ac.za)
nicole L Brooks (brooksn@cput.ac.za)
Omolola R Ayepola (lolaayepola@yahoo.com)

Version: 2
Date: 11 February 2015

Author's response to reviews:

Nutrition and Chronic Diseases Unit,
Oxidative Stress Research Centre
Department of Biomedical Sciences,
Faculty of Health and Wellness,
Cape Peninsula University of Technology,
P.O. Box 1906
Bellville, 7535
South Africa.

9th, February, 2015.

Editor-in-Chief,
BMC Complementary and Alternative Medicine
Dear Sir/Madam,

Please find enclosed the revised version of our manuscript titled: Effects of kolaviron on hepatic oxidative stress in streptozotocin induced diabetes: by Ayepola et al., Find below point by points revisions made to the manuscript.

1. The source of the plant material, voucher number and the person who formally identified the plant material has been included in the methods section.

2. The heading ‘Authors' Contributions’ has been included in the manuscript.
3. The manuscript was revised to include line and page numbers.

Regards,

Omolola R. Ayepola